COMMENTS TO AUTHORS

Dear authors, Your manuscript entitled "Fibrosis in nonalcoholic fatty liver disease: Noninvasive assessment using CT volumetry" has been reviewed. My recommended disposition of the manuscript is to be accepted, despite several limitations of the study that you already mentioned on the Discussion section.

Replay

Thank you very much for your comments.